Author's response to reviews

Title: Analysis on associated factors of uncontrolled hypertension among elderly hypertensive patients in southern China: a community-based, cross-sectional survey

Authors:
- Li Yang Jr (phoebe84331@hotmail.com)
- Xiaoling Xu III (619858830@qq.com)
- Jing Yan III (zjicu@163.com)
- Wei Yu III (yuweipost@163.com)
- Xinhua Tang Dr (tangxh@medmail.com.cn)
- Haibin Wu Jr (hbwu@cdc.zj.cn)
- Christy L. Parkin Dr (clparkin@susdc.org)

Version: 5
Date: 13 August 2014

Author's response to reviews: see over
Dear editor,

Ms. No.: 3702310001212463

Title: Analysis on associated factors of uncontrolled hypertension among elderly hypertensive patients in Southern China: a community-based, cross-sectional survey

Thank you very much for your e-mail on 8th August. We would like to express our sincere thanks to you for the constructive and positive comments. We have revised the manuscript, taking the editor’s and reviewers’ comments into consideration. In particular, point by point responses to the concerns have been listed in details below.

Replies to editor’s comments:

Q1, Delete Funding source heading and move funding information to the Acknowledgements section.

Response:

Thanks for editor’s comment. We deleted the Funding source heading and moved the funding information to the Acknowledgements section according to your suggestion.

Q2, Use authors initials in the Authors’ Contributions section.

Response:

Thanks for editor’s comment. We used authors initials in the Author’s Contributions section according to your suggestion.

Replies to Reviewer:

Q1, Table 2 & table 3: the statistical method for calculating the P-value need to be specified in the endnote.

Response:

Thanks for reviewer’s comment. We added the statistical method for calculating the P-value in the endnotes on Table 2 and table 3.

Q2, Table 4: it will be good if the variables adjusted in the models are detailed in the endnotes.

Response:
Thanks for reviewer’s comment. We added the variables adjusted in the models in the endnote on table 4.

We tried our best to improve the manuscript and made some changes in the manuscript. These changes will not influence the content and framework of the paper. We appreciate for Editors/Reviewers’ warm comments and suggestions earnestly, and hope that the corrections will meet with your approval. Should you have any questions, please contact us without hesitation at the following address, phone and e-mail:

   Address: Center for Cardiovascular Disease Control and Prevention and Zhejiang Hospital, 12 Lingyin Road, Hangzhou 310013, Zhejiang, China.
   Tel.: +86 18268823078, E-mail: phoebe84331@hotmail.com

Once again, thank you very much for your comments and suggestions.

Best regards.

   Li Yang on behalf of the authors